Author’s response to reviews

Title: Longitudinal shortening remains the principal component of left ventricular pumping in patients with chronic myocardial infarction even when the absolute atrioventricular plane displacement is decreased

Authors:

Daniel Asgeirsson (daniel.asgeirsson@gmail.com)
Erik Hedstrom (erik.hedstrom@med.lu.se)
Jonas Jogi (jonas.jogi@med.lu.se)
Ulrika Pahlm (ulrika.pahlm@skane.se)
Katarina Steding-Ehrenborg (katarina.steding-ehrenborg@med.lu.se)
Henrik Engblom (henrik.engblom@med.lu.se)
Hakan Arheden (hakan.arheden@med.lu.se)
Marcus Carlson (marcus.carlsson@med.lu.se)

Version: 2 Date: 20 Jul 2017

Author’s response to reviews:

1. Thank you for including your Authorship Change Form in the manuscript. However, please could you remove this form as supplemental files and email it directly to the Editorial Office (hassee.echanes@biomedcentral.com), so that they can keep it for their records.

We have removed the files and e-mailed them to the Editorial Office.

2. Please can you remove the bullet points from the Declarations sections, and include all the headings for the separate Declarations:
Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations.

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

We have removed the bullet points and updated the Declarations as required.

3. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

This has been added.

4. Please confirm whether the informed consent which obtained from all participants was written or verbal, and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is
deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Written consent was obtained and this information has been added.

5. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Abbreviations have been added.

6. We would also like to ask for you to provide more justification for the contributions of JJö and HA, as currently they do not automatically qualify for authorship. Contribution to data, alone, does not usually justify authorship.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND
d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

The contributions of JJö and HA have been updated. The mistake was made during the revision process when UP was added and the place of JJö and HA moved. This has been corrected and the proper contributions have been added. JJö participated in analysis and interpretation of the data and HA participated in the conception and design of the study.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A clean version has been uploaded.